Author's response to reviews

Title: The role of multipotent cancer associated fibroblasts in hepatocarcinogenesis

Authors:

Caecilia H C Sukowati (caecilia.sukowati@csf.units.it)
Beatrice Anfuso (b.anfuso@csf.units.it)
Lory S Croce (lcroce@units.it)
Claudio Tiribelli (ctliver@csf.units.it)

Version: 2 Date: 13 October 2014

Author's response to reviews:

Dear BMC Cancer staff,

in response to the e-mail sent on 10 October 2014, I had revised the format of the Ms as below:

1. A continuous line numbering on the body of the Ms had been added
2. Section Introduction had been changed to Background

Thank you,

Caecilia Sukowati